Author’s response to reviews

Title: Preliminary application of 3.0T magnetic resonance chemical exchange saturation transfer imaging in brain metastasis of lung cancer

Authors:

Yonggui Yang (yangyonggui125@sina.cn)
Xiaobo QU (quxiaobo@xmu.edu.cn)
Yihui HUANG (huangyihui@stu.xmu.edu.cn)
Gang Guo (guogangxm@163.com)
Shaoyin DUAN (xmdsy@xmu.edu.cn)

Version: 4 Date: 04 Dec 2019

Author’s response to reviews:

Dear

Thank you for giving us the opportunity to revise and resubmit this manuscript. We really appreciate the time and detail provided by each reviewer and by you. The manuscript has certainly benefited from these insightful and revision suggestions.

Editor Comments:

1. Competing interests

Xiaobo QU was blinding to the editorial processes and had no editorial involvement with the peer review of this manuscript.

The authors declare that they have no competing interests.

2. Availability of data and material

All data generated or analyzed during this study are included in this published article.

Best wishes,

Yonggui YANG.